Course Information Sheet, CSC364 session L5101, Summer 2003

This sheet contains most of the important information you need to know about this course. For more detailed information see the course website:

http://www.cs.toronto.edu/~mreza/CSC364

Note: You must read the “Main page” of this website as it contains some important information regarding the course. During the session, announcements will be made at the begining of lectures and will be posted on this website. It is your responsibility to check this site frequently (at least twice a week).

Instructor Information:

ffl Instructor: Mohammad R. Salavatipour ffl E-mail: mreza@cs.toronto.edu ffl Office: SF3208 ffl Phone: (416)978-8700

The best way to contact me is by e-mail. The phone line is shared with my officemates. So PLEASE call me only if it is an emergency.

Time tables:

ffl Lecture: T 6-8, in BA1170 (Extra lecture July 2nd, BA1190, 4-6pm) ffl Tutorial: T 8-9.

Your Last name TA name Room A-F Yilan Gu BA1170 G-L Pixing Zhang BA2135 M-S Spyros Angelopoulos BA2175 T-Z Periklis Papakonstantinou BA1230 ffl Office hour: T 4-5, TBA.

Text book: There is no required text book for this course.

Suggested references: Cormen, Leiserson, Rivest, and Stein: Introduction to Algorithms (Second Edition), MIT Press and McGraw-Hill. The most relevant chapters for this course are: 3, 15, 16, 25, 26, and 34.

M. Sipser: Introduction to the Theory of Computations. Chapters 3,4, and 5. Course notes: Will be posted on the webpage of the course.

Grading Scheme:

ffl Three assignments at 5% each: 15%. ffl Three term tests at 15% each: 45%. ffl Final exam: 40%.

NOTE:

ffl In order to pass the course, you must achieve at least 35% on your final exam, regardless of your other marks.

ffl For any problem in an assignment, term test, or the final exam, you will receive 20% of the mark for that problem if you only write "I don’t know how to answer to this question", and nothing else!!.

Important Dates:

Date Event Tue, May 13 First lecture and first tutorial Wed, June 4 Assignment 1 due Tue, June 10 Term test 1, Assignment 2 posted Tue, July 1 No lecture (University closed) Wed, July 2 Assignment 2 due. Extra lecture in BA1190, 4-6pm Tue, July 8 Term test 2, Assignment 3 posted Sun, July 20 Last day to drop the course Wed, July 23 Assignment 3 due Tue, July 29 Term test 3 Tue, Aug 5 Last lecture

Assignments and Plagiarism:

All assignments are due at 6:00pm in the drop box located on the second floor of Sandford Fleming building. Late assignments are NOT accepted!!.

The work you submit must be your own. You may discuss the assignments with each other, however, you must prepare written solutions alone. On the front page of your assignment you must name the other students you have discussed with, or any other sources you have used. This list will not affect your mark. Copying assignments is a serious academic offence and will be dealt with accordingly.